Author’s response to reviews

Title: A Genotype-specific, randomized-controlled behavioral intervention to improve the neuroemotional outcome of cardiac surgery: study protocol for a randomized controlled trial

Authors:

Daniela Hauer (daniela.hauer@med.uni-muenchen.de)
Iris-Tatjana Kolassa (iris.kolassa@uni-ulm.de)
Rüdiger P Laubender (laubend@ibe.med.uni-muenchen.de)
Ulrich Mansmann (mansmann@ibe.med.uni-muenchen.de)
Christian Hagl (christian.hagl@med.uni-muenchen.de)
Benno Roozendaal (b.roozendaal@cns.umcn.nl)
Dominique JF de Quervain (dominique.dequervain@unibas.ch)
Gustav Schelling (gustav.schelling@med.uni-muenchen.de)

Version: 3 Date: 10 March 2013

Author’s response to reviews: see over
Cover Letter

We have made the changes to the title as requested:

It has now been changed to “A Genotype-specific, randomized-controlled behavioral intervention to improve the neuroemotional outcome of cardiac surgery: study protocol for a randomized controlled trial”.

Sincerely,

Daniela Hauer

First author